

GOPEN ACCESS

Citation: López-Rosero A, Sippy R, Stewart-Ibarra AM, Ryan SJ, Mordecai E, Heras F, et al. (2024) High prevalence of Zika virus infection in populations of *Aedes aegypti* from South-western Ecuador. PLoS Negl Trop Dis 18(1): e0011908. https://doi.org/10.1371/journal.pntd.0011908

Editor: Mariangela Bonizzoni, Universita degli Studi di Pavia, ITALY

Received: July 31, 2023

Accepted: January 8, 2024

Published: January 18, 2024

Peer Review History: PLOS recognizes the benefits of transparency in the peer review process; therefore, we enable the publication of all of the content of peer review and author responses alongside final, published articles. The editorial history of this article is available here: https://doi.org/10.1371/journal.pntd.0011908

Copyright: © 2024 López-Rosero et al. This is an open access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: All relevant data are within the paper and its <u>Supporting Information</u> files.

RESEARCH ARTICLE

High prevalence of Zika virus infection in populations of *Aedes aegypti* from Southwestern Ecuador

Andrea López-Rosero¹, Rachel Sippy^{2¤a}, Anna M. Stewart-Ibarra^{2¤b}, Sadie J. Ryan³, Erin Mordecai⁴, Froilán Heras², Efraín Beltrán⁵, Jaime A. Costales¹, Marco Neira₁₀^{1¤c}*

 Centro de Investigación para la Salud en América Latina (CISeAL), Facultad de Ciencias Exactas y Naturales, Pontificia Universidad Católica del Ecuador, Quito, Ecuador, 2 Institute for Global Health and Translational Science and Department of Medicine, SUNY Upstate Medical University, Syracuse, New York, United States of America, 3 Quantitative Disease Ecology and Conservation (QDEC) Laboratory, Department of Geography and Emerging Pathogens Institute, University of Florida, Gainesville, Florida, United States of America, 4 Department of Biology, Stanford University, Stanford, California, United States of America, 5 Universidad Técnica de Machala, Machala, El Oro, Ecuador

¤a Current address: School of Mathematics and Statistics, University of St. Andrews, Fife, United Kingdom
¤b Current address: Inter-American Institute for Global Change Research, Montevideo, Uruguay
¤c Current address: Climate and Atmosphere Research Center (CARE-C), The Cyprus Institute, Nicosia, Cyprus

* m.neira@cyi.ac.cy

Abstract

We performed an arboviral survey in mosquitoes from four endemic Ecuadorian cities (Huaquillas, Machala, Portovelo and Zaruma) during the epidemic period 2016–2018. Collections were performed during the pre-rainy season (2016), peak transmission season (2017) and post-rainy season (2018). *Ae. aegypti* mosquitoes were pooled by date, location and sex. Pools were screened by RT-PCR for the presence of ZIKV RNA, and infection rates (IRs) per 1,000 specimens were calculated. A total of 2,592 pools (comprising 6,197 mosquitoes) were screened. Our results reveal high IRs in all cities and periods sampled. Overall IRs among female mosquitoes were highest in Machala (89.2), followed by Portovelo (66.4), Zaruma (47.4) and Huaquillas (41.9). Among male mosquitoes, overall IRs were highest in Machala (35.6), followed by Portovelo (33.1), Huaquillas (31.9) and Zaruma (27.9), suggesting that alternative transmission routes (vertical/venereal) can play important roles for ZIKV maintenance in the vector population of these areas. Additionally, we propose that the stabilization of ZIKV vertical transmission in the vector population could help explain the presence of high IRs in field-caught mosquitoes during inter-epidemic periods.

Author summary

Zika virus (ZIKV) is a mosquito-borne virus native to Africa that can cause a wide range of symptoms in humans, including neurological complications and fetal abnormalities. ZIKV arrived to South America in 2015, and quickly spread through the continent. In Ecuador, this virus caused several thousand infections during 2016 and 2017, but its **Funding:** RS, AMSI, SJR, MN received funding from the National Science Foundation Zika Rapid (DEB-161145). ALR, RS, AMSI, SJR, EM, FH, MN received funding from the National Science Foundation Ecology and Evolution of Infectious Diseases grant (DEB-52 1518681). EM received support from the U.S. National Science Foundation (DEB-2011147, with the Fogarty International Center), the National Institutes of Health (R35GM133439), and the Stanford University Woods Institute for the Environment and Center for Innovation in Global Health. The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

Competing interests: The authors have declared that no competing interests exist.

incidence dropped suddenly thereafter, with just 11 cases reported between 2018 and 2022. The reasons behind this abrupt decrease are not well understood.

In this work, we performed an arboviral survey in mosquitoes from four endemic Ecuadorian cities during the period 2016–2018. We screened 2,592 pools (comprising 6,197 specimens) for the presence of ZIKV, and estimated infection rates in female and male mosquitoes. We found high infection rates in all cities and periods sampled. Considering the low incidence of human cases during 2018, our results suggest that in our study areas the virus could be maintained by mechanisms that either do not depend on human infection, or cause low rates of symptomatic infections. High rates of ZIKV infection in male mosquitoes found during our study suggest that vertical and/or venereal transmission routes could play important roles in the inter-epidemic maintenance of this pathogen in our study area.

Introduction

Arthropod-borne viruses, also known as arboviruses, are usually transmitted between hosts by the bite of blood-feeding arthropods, including mosquitoes, sandflies and ticks [1]. A subgroup of arboviruses, which include dengue virus (DENV), Zika virus (ZIKV) and chikungunya virus (CHIKV), are preferentially transmitted by *Aedes aegypti* and *Ae. albopictus* mosquitoes. Because these mosquito species are highly synanthropic, *Aedes*-borne viruses are spreading quickly and have become major threats to global public health [2].

ZIKV is a member of the *Orthoflavivirus* genus, which belongs to the *Flaviviridae* family of viruses. Virions are 40–60 nm in diameter and contain a single-stranded 11-kb RNA genome, which encodes three structural and seven non-structural proteins [3,4]. Under sylvatic conditions, ZIKV circulates between non-human primates and various species of *Aedes* mosquitoes. However, humans exposed to sylvatic environments can become infected and carry the virus into urban areas, where epidemic cycles can then be maintained between susceptible humans and synanthropic *Aedes* species, principally *Ae. aegypti* and *Ae. albopictus* [4].

ZIKV infection in humans is estimated to be asymptomatic in 80% of cases [5]. When present, symptoms might include rash, low-grade fever, joint and muscle pain, periarticular edema and conjunctivitis [6,7]. A small fraction of infected individuals develops neurological complications, including Guillain-Barré syndrome, meningoencephalitis and myelitis [8,9]. ZIKV infection in pregnant women has been linked to serious complications including fetal loss, congenital microcephaly, ventriculomegalia and eye abnormalities, among others [10,11].

First isolated in Uganda in 1947, ZIKV was first reported have spread outside of Africa in the 1960s, when it was detected in Asia. For the next two decades, sporadic cases of ZIKV infection were reported within Africa and Asia, until 2007, when the virus caused an outbreak in Micronesia. In 2013 and 2014 the virus spread to French Polynesia and Easter Island, reaching Brazil in 2015 [7,12]. Within a year, ZIKV caused a massive outbreak that spread throughout the Americas, leading to the World Health Organization to declare this disease a public health emergency of international concern in 2016 [13]. In 2017, the first cases of ZIKV were reported in India [4].

With cases being widely under-reported and the dynamics of ZIKV transmission still unclear, it is critical to measure virus circulation in mosquito vector populations over time, to characterize transmission risk and modes. In this study, we aim to characterize ZIKV infection rates in vectors from Ecuador during the epidemic period (2016–2018) to understand the extent of transmission risk and the frequency of vertical transmission (via viral detection in

	2015	2016	2017	2018	2019	2020	2021	2022
Dengue	42,473	14,159	11,387	3,099	8,416	19,950	20,592	16,017
Zika	1	2,902	2,331	9	0	0	2	0

Table 1. Number of dengue and Zika cases in Ecuador, 2015–2022. Data sources: [14,15].

https://doi.org/10.1371/journal.pntd.0011908.t001

non-biting male mosquitoes). Zika dynamics followed a sharp epidemic curve in Ecuador (Table 1). After appearing in Ecuador in 2015, Zika cases increased rapidly, with 2,947 and 2,423 cases reported during 2016 and 2017, respectively (Table 1). However, incidence dropped suddenly to just 9 cases in 2018, and only two new cases have been reported since that year [14] (Table 1). In this study, we screened male and female *Ae. aegypti* mosquitoes collected in the southern province of El Oro for the presence of Zika virus. We calculate infection rates for each sex, city and time period, and discuss potential biological and epidemiological implications of our data.

Methods

Ethics statement

Before the initiation of field activities, the protocols used in this study were dutifully approved by Institutional Review Boards in Ecuador (Luis Vernaza Hospital and Ministry of Health) and the USA (State University of New York Upstate Medical University, IRBNET ID 4177710–25). Written informed consent was obtained from all adult participants (18 years of age or older).

Study area

Samples were collected in four cities (Machala, Huaquillas, Portovelo and Zaruma) of El Oro province, which is located in southwestern Ecuador, and borders Peru (Fig 1 and Table 2). Although all four cities present endemic transmission of *Aedes*-borne arboviruses (specifically, dengue virus), they differ in the incidence of arboviral disease: while Machala and Huaquillas present high numbers of cases every year, higher-elevation Portovelo and Zaruma present more limited incidence [16].

Sampling period

The study was conducted between 2016 and 2018, across different seasons in different years to capture a fuller range of conditions. Peak mosquito abundance and arbovirus transmission occurs in the area during the rainy season, which takes place from December to May, and transmission persists at low levels during the rest of the year [18]. During 2016, the collection of mosquitoes was performed between July and December, corresponding to the pre-rainy season. In 2017, collections took place between January and May, which corresponds to the peak transmission season and, in 2018, collections were performed between April and August, corresponding to the post-rainy period.

Mosquito collection, pooling and RNA extraction

Each city was geographically divided in 10 clusters. In the center of each cluster, one house was selected for participation, and additional houses (up to 5 per cluster, within a distance of 250 meters from the central house) were randomly enrolled into the study. Houses with air conditioning were excluded. Participating houses were georeferenced and issued a cluster and house identification code. Mosquito collections were performed every two weeks in these houses,



Fig 1. Study area. The country map on the right shows the location of El Oro province (dark green area) within Ecuador (light green area). Inset on the left corresponds to the area shaded in pink on the country map. The cities where this study took place are marked with orange dots and labelled on the inset. Areas colored in grey correspond to neighboring countries. Basemap source: OpenStreetMap (https://openstreetmap.org). Administrative boundaries shapefile source: The Humanitarian Data Exchange (https://data.humdata.org/dataset/cod-ab-ecu).

https://doi.org/10.1371/journal.pntd.0011908.g001

including all indoor spaces as well as porch/patio spaces. If residents were not home during a given collection period, the house was skipped for the collection period. Mosquitoes were collected using manual Prokopack aspirators [19]. Following each collection, mosquitoes were transported on ice to a field laboratory where *Ae. aegypti* specimens were identified and pooled by sex and cluster in groups of ≤ 10 specimens per house for further processing.

For homogenization, each pool was placed in a 2 mL sterile microcentrifuge tube containing 1mL TRI Reagent (Molecular Research Center, Inc., Cincinnati, OH, USA) and two small, sterile metal beads. Tubes were capped tightly and vortexed until no whole abdomens were visible. The homogenate was transferred to a new sterile microcentrifuge tube and stored at -20°C until transported to the laboratories at the *Centro de Investigación para la Salud en América Latina*' (CISeAL), in Quito, where samples were further processed.

City	Location	Altitude	Population	Arboviral transmission	Description
Machala	3°15'09"S, 79°57'20"W	6m	~289,000	High	Coastal city. Province capital with intense agricultural and commercial activity.
Huaquillas	3°28'33"S, 80°13'33"W	15m	~60,000	High	Coastal city. Borders Peru. Intensive cross-border commerce and trade.
Portovelo	3°42'58"S, 79°37'08"W	645m	~14,000	Limited	Inland. Intense mining activity.
Zaruma	3°41'31"S, 79°36'47"W	1,155m	~26,000	Limited	Inland. Intense mining activity.

Table 2. Characteristics of cities where collections were performed. Population data rounded from [17]. Arboviral transmission data from [16].

https://doi.org/10.1371/journal.pntd.0011908.t002

Total RNA was extracted following standard TRI Reagent manufacturer's recommendations. RNA pellets were washed in 75% EtOH, re-suspended in 50 μ l of nuclease-free water and stored at -80°C until further processing.

cDNA synthesis

cDNA synthesis was performed using the GoScript Reverse Transcription Kit (Promega Corp., Wisconsin, USA), following manufacturer's recommendations. Briefly, 2 μ l total RNA were mixed with 1 μ L random primers (0.5 μ g/ μ L) and 2 μ l nuclease-free water. The mixture was heated to 70°C for 5 minutes to linearize RNA and then cooled to 4°C for 5 minutes. Linearized RNA was then mixed with 15 μ l of reverse transcription mix (containing 4 μ l 5X GoScript buffer, 2 μ l MgCl₂ 25nM, 1 μ l 10 mM dNTPs and 1 μ l GoScript reverse transcriptase). This cocktail was subsequently incubated at 25°C for 5 minutes, 42°C for 1 hour and 70°C for 15 min. Synthetized cDNA was stored at -80°C until further use.

RNA extraction / cDNA synthesis control

In order to verify the success of the RNA extraction and cDNA synthesis, each sample was screened for the presence of mRNA corresponding to the housekeeping gene *Actin-1*, following the protocol described by [20]. Only samples for which *Actin-1* mRNA was successfully amplified were subsequently screened for arbovirus presence.

Negative control reactions

In order to control for contamination in the reagents, each batch of PCR reactions contained one reaction where the cDNA template was replaced by 1 μ L of sterile water.

Polymerase chain reaction (PCR)

Samples were initially screened for the presence of Flaviviral RNA with the primer set MA/ CFD2, originally described by Kuno [21] (Table 3). PCR was performed using the GoTaq Flexi DNA polymerase system (Promega Corp., Wisconsin, USA), following manufacturer's recommendations. Briefly, each reaction contained 1.25 units GoTaq DNA Polymerase, 1X Green GoTaq Flexi Buffer, 2 mM MgCl₂ Solution, 0.2 mM dNTP mix, each primer at 1 μ M concentration and 1 μ L cDNA as template. Amplification was performed by subjecting samples to 95°C for 2 minutes, followed by 35 cycles of 95°C for 30 seconds (denaturation), 53°C for 30 seconds (annealing), and 72°C for 1 minute (extension), with a final extension step of 72°C for 5 minutes. All PCR products were visualized on 1.5% agarose gels in TBE buffer (Tris-base, Boric acid, 0.5 M EDTA at pH 8.0), containing ethidium bromide.

Samples generating positive results for the presence of *Orthoflavivirus* RNA were tested for the presence of ZIKV using a nested-PCR strategy. In a first round of amplification, we

Primer pair	Primer source	Forward primer name (and sequence)	Reverse primer name (and sequence)	Expected amplicon size
MA/CFD2	Kuno, 1998 [21]	MA (5'-CATGATGGGRAARAGRGARRAG-3')	CFD2 (5'- GTGTCCCAGCCGGCGGTGTCATCAGC-3')	260bp
ZIKENV	Faye et al., 2008 [22]	ZIKVENVF (5'-GCTGGDGCRGACACHGGRACT-3')	ZIKVENVR (5'-RTCYACYGCCATYTGGRCTG-3')	364bp
ZIKV-NES1	Custom-designed	ZIKV-nes1-F (5'-AAACCGTCGTCGTTTCTGGG-3')	ZIKV-nes1-R (5'-TGCATACTGCACCTCCACTG-3')	242bp

Table 3. Primers used for screening procedure.

https://doi.org/10.1371/journal.pntd.0011908.t003

screened each sample with the primer set ZIKENV (Table 3), originally described by Faye et al. [22], which amplifies a 364 bp segment of the envelope protein coding region of the ZIKV genome (accession AY632535). Reagent concentrations and thermal profile were the same as previously described, excepting that annealing temperature for amplification with primer set ZIKENV was set to 55°C.

Amplicons of the right size obtained during the first amplification were gel-purified using the Wizard SV Gel and PCR Clean-Up System (Promega Corp., Wisconsin, USA). Five µL purified amplicon were used as template for a second round of amplification using primer set ZIK-NES1 (Table 3), which was custom-designed to target a 242bp portion of the amplicon obtained with primer pair ZIKENV. Other reagent concentrations and thermal profile were the same as previously described, excepting that annealing temperature for amplification with primer set ZIK-NES1 was set to 50°C.

Sequencing

Amplicons of the expected size obtained after the second round of amplification were gel-purified and sent to a commercial provider (sequencing facilities at 'Universidad de las Americas' in Quito, Ecuador) for Sanger-sequencing. Sequence identity was confirmed by querying amplicon sequences against information available at the U.S. National Center for Biotechnology Information using their 'Basic Local Alignment Search Tool' (BLAST) [23].

Infection rate (IR) estimation

IRs were calculated using the 'PooledInfRate' package, a Microsoft Office Excel add-In to compute prevalence estimates from pooled samples [24], following the software developer's recommendations. Bias-corrected maximum likelihood estimate (MLE) was chosen as point estimate because this calculation does not assume that when a pool tests positive for an arbovirus, only one individual in said pool is positive (an assumption required for the calculation of the more traditional 'minimum infection rate') [24]. All IR presented in this report correspond to values per 1,000 individuals.

Results

A total of 6,197 mosquitoes were collected during this work. These were grouped in 2,592 pools, corresponding to 1,355 pools of female specimens, and 1,237 pools of male specimens (average pool size for each sex, region and year are shown in <u>S1 Table</u>). As expected, the highest abundance of both female and male pools in all four cities was observed during the rainy season (2017), followed by the post-rainy season (2018). Collections during the pre-rainy season (2016) yielded the lowest abundance of mosquito pools (Fig 2).

Only nine pools did not generate the expected results in the positive control reactions for the detection of *Actin-1* mRNA, and were therefore excluded from further analysis. The remaining 2,583 pools (corresponding to 99.6% of the total) generated the expected amplicon in the positive control reaction, indicating successful RNA extraction and cDNA synthesis. These samples were further screened for the presence of arboviral RNA.

Overall, 277 pools tested positive for ZIKV, representing 10.7% of the total number of pools screened. This corresponded to an IR of 46.9 per 1,000 individuals. The distribution of these results for each city and year is shown in Tables <u>4</u> and <u>5</u> and in Fig <u>3</u>. A sub-set of the partial ZIKV sequences obtained were deposited in GenBank (www.ncbi.nlm.nih.gov/genbank/), accession numbers OM831146, OM831147, OM831148, OM831149, OM831150, OM831151, OM831152, OM831153 and OM831154.





https://doi.org/10.1371/journal.pntd.0011908.g002

We detected ZIKV infection in mosquitoes of both sexes in all cities and almost all sampling periods (the only exception being male pools in the city of Portovelo during 2016). All four cities presented the highest infection rates during the 2018 season, the year in which we sampled in the post-rainy season, in both female and male pools. In fact, all cities except Zaruma showed a continuous increase in infection rates in both sexes each year throughout the sampling period.

City	Year	No. pools		No. Positive	No. Positive		% positive	
		female	male	female	male	female	male	
Huaquillas	2016	57	46	1	2	1.8	4.3	
	2017	285	224	25	14	8.8	6.3	
	2018	124	123	24	17	19.4	13.8	
	City sub-total	466	393	50	33	10.7	8.4	
Machala	2016	80	71	4	2	5.0	2.8	
	2017	223	219	23	10	10.3	4.6	
	2018	143	127	50	25	35.0	19.7	
	City sub-total	446	417	77	37	17.3	8.9	
Portovelo	2016	22	21	1	0	4.5	0.0	
	2017	143	147	12	6	8.4	4.1	
	2018	102	90	22	11	21.6	12.2	
	City sub-total	267	258	35	17	13.1	6.6	
Zaruma	2016	29	25	3	1	10.3	4.0	
	2017	86	93	4	1	4.7	1.1	
	2018	56	48	11	8	19.6	16.7	
	City sub-total	171	166	18	10	10.5	6.0	

Table 4. Total number of pools of adult male and female Ae. aegypti tested (and positive for ZIKV) for each city and year.

https://doi.org/10.1371/journal.pntd.0011908.t004

City	Year	Female			Male	Male		
		IR	95% CI—lower limit	95% CI—upper limit	IR	95% CI—lower limit	95% CI—upper limit	
Huaquillas	2016	9.4	0.5	44.6	18.9	3.5	59.5	
	2017	29.9	20.0	43.0	20.0	11.5	32.5	
	2018	95.2	64.7	134.5	73.5	44.9	113.7	
	Overall	41.9	31.8	54.2	31.9	22.5	43.9	
Machala	2016	27.6	9.1	64.5	11.2	2.0	36.1	
	2017	45.0	29.6	65.6	15.2	7.8	26.9	
	2018	240.9	187.4	302.1	122.1	81.4	175.8	
	Overall	89.2	72.0	109.2	35.6	25.6	48.2	
Portovelo	2016	35.4	2.1	157.9	0.0	0.0	120.5	
	2017	43.1	23.7	71.9	19.1	7.9	39.3	
	2018	99.1	66.6	141.2	63.1	33.7	108.1	
	Overall	66.4	47.9	89.6	33.1	20.0	51.5	
Zaruma	2016	67.3	18.5	168.2	22.7	1.3	103.7	
	2017	18.2	6.0	42.9	4.2	0.2	20.3	
	2018	94.0	51.5	156.9	103.2	48.7	192.0	
	Overall	47.4	29.5	72.0	27.9	14.4	49.2	

Table 5. Infection rates (MLE per 1,000 specimens) of ZIKV in adult Ae. aegypti pools for each city and year.

https://doi.org/10.1371/journal.pntd.0011908.t005





https://doi.org/10.1371/journal.pntd.0011908.g003

In female mosquitoes, IRs varied over 25-fold, between 9.4 (Huaquillas, 2016) and 240.9 (Machala, 2018). Overall (i.e. all years combined) infection rates among female mosquitoes were highest in Machala (89.2), followed by Portovelo (66.4), Zaruma (47.4) and Huaquillas (41.9).

IRs among male mosquitoes were generally lower than among females, varying between 0 (Portovelo, 2016) and 122.1 (Machala, 2018). Overall infection rates in male mosquitoes were highest in Machala (35.6), followed by Portovelo (33.1), Huaquillas (31.9) and Zaruma (27.9).

Discussion

The epidemiology of ZIKV in Ecuador remains poorly understood. After arriving in the country sometime during 2015 (as evidenced by the single human case reported for that year) (Table 1), the virus spread quickly, causing over 2,000 cases per year during 2016 and 2017, and then dropped suddenly to just 9 cases in 2018 (Fig 1) [14]. Since 2019 until the present day, only two cases have been reported in Ecuador, both in 2021 [14,25]. The epidemiological trend in neighboring countries has been similar: Colombia saw a steady decline in the number of new Zika cases, going from 91,711 in 2016 to 138 in 2022. And Peru went from 5,361 new cases in 2017, to just 20 cases in 2022 [25].

The virtual disappearance of Zika from Ecuador is in stark contrast with the epidemiological trend of dengue virus, another *Orthoflavivirus* transmitted mainly by the same vector (*Ae. aegypti*) in this country. Dengue has been consistently present in Ecuador over the last three decades [26], with a yearly incidence that, during the last five years, has ranged from approximately 3,000 cases in 2018, to over 20,000 cases in 2021 (Table 1) [14,15]. This suggests that the drastic reduction in Zika incidence observed since 2018 is not the result of a corresponding reduction in vector population, nor does it represent a reduction in the frequency of contact between vectors and human hosts. Therefore, we must look into other elements of the transmission cycle in order to shed some light into this puzzling situation.

Former reports have revealed ZIKV IRs as high as 53/1,000 [27] among female *Ae. aegypti* from provinces of the central and northern Ecuadorian coast in 2016 [27]. In agreement, our data confirms that high IRs in the *Ae. aegypti* populations were maintained throughout the study period in the southern coastal province of El Oro. Additionally, our data shows that IR in mosquitoes from all four cities progressively increased between 2016 and 2018, in spite of the fact that the number of reported Zika cases in the entire province actually declined between 2017 and 2018 (66 and 0 cases, respectively) [28]. This suggests that in our study areas, the virus is maintained by mechanisms that either do not depend on human infection, or cause extremely low rates of symptomatic infection in humans.

Although the nature of our data does not allow us to elucidate the exact nature of such mechanisms, one possibility is the maintenance of ZIKV in the mosquito populations through vertical and/or venereal transmission. Previous experimental work found high rates of ZIKV venereal transmission in *Ae. aegypti* (both male-to-female and female-to-male) [29], providing evidence that this route constitutes a mechanism for virus dissemination among the vector population. Furthermore, vertical transmission of ZIKV in *Ae. aegypti* has been reported to occur under both laboratory and natural conditions [30–32].

Because male mosquitoes can only acquire arboviral infections via either vertical or venereal route, presence of infected males in the field is generally considered as a marker of vertical and/or venereal transmission occurring in the area [33]. Interestingly, our data shows high infection rates in male as well as female mosquitoes, suggesting that these alternative transmission routes contribute to the maintenance of ZIKV in our study region. This hypothesis would be consistent with previous reports from neighboring Colombia, which found that vertical/ venereal transmission pathways play an important role for the maintenance of this virus during inter-epidemic periods [33].

Mosquitoes of the *Aedes* genus display higher rates of vertical transmission than other genera [34]. Accordingly, high rates of ZIKV vertical transmission in *Ae. aegypti* have been previously reported [35,36]. Ciota and collaborators [36] found IR as high as 28.5 among the offspring of experimentally infected female mosquitoes from South-American populations, and have stated that different *Ae. aegypti* populations might vary substantially on their ability to transmit ZIKV vertically. Although the IR among male specimens in our study could reach substantially higher values in specific populations and years, the overall (i.e. pooled for all years) IR in males for each city varies between 27.9 and 35.6 (Table 5), which is consistent with values presented in previous reports [33,36]. It is also worth noticing that due to our sampling methodology, infected mosquitoes in different pools could potentially originate from the same infected parent, thereby contributing to the relatively high IR observed in both male and female groups.

According to Reese et al. [37], wild mosquito populations in which vertical transmission has stabilized (i.e. females consistently transmit arbovirus to a high percentage of their offspring), show a high proportion of transovarially-infected individuals generating non-productive infections (i.e. infections in which no infective virions can be detected, in spite of the presence of viral antigens and nucleic acids). The study in question found that, at one of their study sites, 7 out of 58 tested specimens (equivalent to 120.7/1,000) displayed such non-productive infections. Transovarially-infected mosquitoes generating infective virions were >10 fold less abundant in these populations. If we hypothesize that a similar situation could be occurring with ZIKV and Ae. aegypti in our study area, this could help explain, at least in part, the low number of human infections observed during 2018 in spite of the high IR observed in our molecular screening process, as a significant proportion of mosquitoes testing positive for the presence of arboviral RNA would in fact represent non-productive infections, and would therefore play no role in the transmission of ZIKV to humans. Furthermore, this 'stabilization hypothesis' could also help explain the progressive increase in IR observed throughout our study period, as the proportion of vertically-infected specimens would be expected to increase over time. Efficient venereal transmission among adult mosquitoes could further contribute to progressive increases in IR.

To test this stabilization hypothesis, it would be necessary to evaluate the viability of viral particles obtained from field-caught mosquitoes. Unfortunately, this was not part of our original study design, and our laboratory protocol rendered all viral particles inactive at the very early steps of sample processing, making it impossible for us to quantify the proportion of productive vs non-productive infections. Therefore, we recommend that future studies in the area incorporate protocols aimed at testing the viability of virus retrieved from the vector populations. To the best of our knowledge, no further information is available regarding the prevalence of ZIKV in the mosquito populations of our study regions after 2018.

The development of herd immunity in the human population is another factor expected to contribute to the decline of ZIKV incidence in the Americas. It has been proposed that following explosive outbreaks in immunologically naive populations, large proportions of said populations can seroconvert within a relatively short time [38]. As an example, a survey conducted in northeastern Brazil found that over 60% of the local population presented serological evidence of anti-ZIKV antibody production within approximately one year of the start of the 2015 outbreak [39]. Unfortunately, no data regarding the prevalence of anti-ZIKV antibodies in the population of our study area is currently available in the scientific literature, making it difficult to estimate the extent to which herd immunity plays a role in limiting transmission in this region.

Further research is required in order to obtain a complete picture of the epidemiological landscape of ZIKV in this region. Continued monitoring of infection in field-caught mosquitoes would provide information on whether ZIKV did in fact become stabilized in these populations, while testing the viability of virus recovered from field-caught mosquitoes would help us confirm (and quantify) the existence of non-productive infections. Furthermore, performing sero-surveys aimed at identifying the prevalence of protective antibodies in the human population would allow us to identify the presence of asymptomatic infections, and estimate the protective effect of herd immunity in these areas.

To the best of our knowledge, this is the first report of vertical transmission of an arbovirus in Ecuador. Although our results cannot fully explain the almost total disappearance of clinical Zika cases in Ecuador after 2018, they provide a glimpse into the complex and dynamic ecologic interactions between arboviruses, arboviral vectors and hosts in nature.

Supporting information

S1 Table. Average mosquito pool size for each sex, region and year. (DOCX)

Acknowledgments

The authors would like to thank Naveed Heydari for his support with sample collection.

Author Contributions

Conceptualization: Anna M. Stewart-Ibarra, Sadie J. Ryan, Erin Mordecai, Marco Neira.

Data curation: Andrea López-Rosero, Rachel Sippy, Froilán Heras, Marco Neira.

Formal analysis: Andrea López-Rosero, Marco Neira.

Funding acquisition: Anna M. Stewart-Ibarra, Sadie J. Ryan, Erin Mordecai, Marco Neira.

Investigation: Andrea López-Rosero, Rachel Sippy, Anna M. Stewart-Ibarra, Froilán Heras, Marco Neira.

Methodology: Andrea López-Rosero, Rachel Sippy, Anna M. Stewart-Ibarra, Froilán Heras, Marco Neira.

Project administration: Rachel Sippy, Anna M. Stewart-Ibarra, Froilán Heras, Marco Neira.

Resources: Anna M. Stewart-Ibarra, Efraín Beltrán, Marco Neira.

Supervision: Rachel Sippy, Anna M. Stewart-Ibarra, Efraín Beltrán, Marco Neira.

Validation: Marco Neira.

Writing - original draft: Andrea López-Rosero, Marco Neira.

Writing – review & editing: Andrea López-Rosero, Rachel Sippy, Anna M. Stewart-Ibarra, Sadie J. Ryan, Erin Mordecai, Froilán Heras, Efraín Beltrán, Jaime A. Costales, Marco Neira.

References

- 1. Hanley KA. Origin and Evolution of Viruses. Amsterdam, The Netherlands: Elsevier; 1998.
- Ogunlade ST, Meehan MT, Adekunle AI, Rojas DP, Adegboye OA, McBryde ES. A review: Aedesborne arboviral infections, controls and wolbachia-based strategies. Vol. 9, Vaccines. MDPI AG; 2021. p. 1–23.

- Hung S-J, Huang S-W. Contributions of Genetic Evolution to Zika Virus Emergence. Front Microbiol. 2021 May 6; 12. https://doi.org/10.3389/fmicb.2021.655065 PMID: 34025610
- Sharma V, Sharma M, Dhull D, Sharma Y, Kaushik S, Kaushik S. Zika virus: An emerging challenge to public health worldwide. Vol. 66, Canadian Journal of Microbiology. Canadian Science Publishing; 2020. p. 87–98.
- Duffy MR, Chen T-H, Hancock WT, Powers AM, Kool JL, Lanciotti RS, et al. Zika Virus Outbreak on Yap Island, Federated States of Micronesia. N Engl J Med. 2009 Jun 11; 360(24):2536–43. https://doi. org/10.1056/NEJMoa0805715 PMID: 19516034
- Musso D, Gubler DJ. Zika virus. Clin Microbiol Rev. 2016 Jul 1; 29(3):487–524. <u>https://doi.org/10.1128/ CMR.00072-15 PMID: 27029595</u>
- 7. Masmejan S, Musso D, Vouga M, Pomar L, Dashraath P, Stojanov M, et al. Zika virus. Vol. 9, Pathogens. MDPI AG; 2020. p. 1–14.
- Brito Ferreira ML, Militão de Albuquerque M de FP, de Brito CAA, de Oliveira França RF, Porto Moreira ÁJ, de Morais Machado MÍ, et al. Neurological disease in adults with Zika and chikungunya virus infection in Northeast Brazil: a prospective observational study. Lancet Neurol. 2020 Oct 1; 19(10):826–39. https://doi.org/10.1016/S1474-4422(20)30232-5 PMID: 32949543
- Cao-Lormeau VM, Blake A, Mons S, Lastère S, Roche C, Vanhomwegen J, et al. Guillain-Barré Syndrome outbreak associated with Zika virus infection in French Polynesia: A case-control study. Lancet. 2016 Apr 9; 387(10027):1531–9.
- de Araújo TVB, Rodrigues LC, de Alencar Ximenes RA, de Barros Miranda-Filho D, Montarroyos UR, de Melo APL, et al. Association between Zika virus infection and microcephaly in Brazil, January to May, 2016: preliminary report of a case-control study. Lancet Infect Dis. 2016 Dec 1; 16(12):1356–63. https://doi.org/10.1016/S1473-3099(16)30318-8 PMID: 27641777
- Pomar L, Musso D, Malinger G, Vouga M, Panchaud A, Baud D. Zika virus during pregnancy: From maternal exposure to congenital Zika virus syndrome. Vol. 39, Prenatal Diagnosis. John Wiley and Sons Ltd; 2019. p. 420–30.
- Yun SI, Lee YM. Zika virus: An emerging flavivirus. Vol. 55, Journal of Microbiology. Microbiological Society of Korea; 2017. p. 204–19.
- 13. Gulland A. Zika virus is a global public health emergency, declares WHO. Vol. 352, BMJ (Clinical research ed.). British Medical Journal Publishing Group; 2016. p. i657.
- Ministerio de Salud Pública del Ecuador. Enfermedades Transmitidas Por Vectores Ecuador, SE 01– 53–2020 [Internet]. 2021. Available from: <u>https://www.salud.gob.ec/wp-content/uploads/2021/01/</u> Vectores-SE-53.pdf.
- Pan American Health Organization. PLISA—Health Information Platform for the Americas [Internet]. 2023 [cited 2023 Jul 1]. Available from: https://www3.paho.org/data/index.php/en/.
- 16. Ryan SJ, Mundis SJ, Aguirre A, Lippi CA, Beltrán E, Heras F, et al. Seasonal and geographic variation in insecticide resistance in *Aedes aegypti* in southern Ecuador. PLoS Negl Trop Dis. 2019 Jun 1; 13(6): e0007448.
- INEC Ecuador. Proyecciones Poblacionales [Internet]. [cited 2022 Jan 13]. Available from: https://www. ecuadorencifras.gob.ec/proyecciones-poblacionales/.
- Stewart Ibarra AM, Ryan SJ, Beltrán E, Mejía R, Silva M, Muñoz Á. Dengue vector dynamics (*Aedes aegypti*) influenced by climate and social factors in Ecuador: Implications for targeted control. PLoS One. 2013 Nov 12; 8(11):e78263.
- Vazquez-Prokopec GM, Galvin WA, Kelly R, Kitron U. A new, cost-effective, battery-powered aspirator for adult mosquito collections. J Med Entomol. 2009 Nov; 46(6):1256–9. https://doi.org/10.1603/033. 046.0602 PMID: 19960668
- Staley M, Dorman KS, Bartholomay LC, Farfan-ale JA, Loroño-pino MA, Julian E, et al. Sequence analysis of actin-1 from diverse mosquito species. 2010; 26(2):214–8.
- Kuno G. Universal diagnostic RT-PCR protocol for arboviruses. J Virol Methods. 1998 May; 72(1):27– 41. https://doi.org/10.1016/s0166-0934(98)00003-2 PMID: 9672130
- 22. Faye O, Faye O, Dupressoir A, Weidmann M, Ndiaye M, Alpha Sall A. One-step RT-PCR for detection of Zika virus. J Clin Virol. 2008 Sep; 43(1):96–101. <u>https://doi.org/10.1016/j.jcv.2008.05.005</u> PMID: 18674965
- National Center for Biotechnology Information (NCBI). Nucleotide BLAST: Search nucleotide databases using a nucleotide query. Basic Local Alignment Search Tool. 2009;3–6.
- Biggerstaff B. PooledInfRate, Version 4.0: a Microsoft Office Excel Add-In to compute prevalence estimates from pooled samples. Fort Collins, CO, U.S.A: Centers for Disease Control and Prevention; 2009.

- PAHO/WHO. Zika—Weekly report [Internet]. 2023 [cited 2023 May 29]. Available from: https://www3. paho.org/data/index.php/en/mnu-topics/zika-weekly-en/.
- Ministerio de Salud Pública del Ecuador. Ecuador en alerta para prevenir el contagio del dengue [Internet]. 2020 [cited 2021 May 7]. Available from: https://www.salud.gob.ec/estrategia-nacional-de-controldel-dengue/.
- Cevallos V, Ponce P, Waggoner JJ, Pinsky BA, Coloma J, Quiroga C, et al. Zika and Chikungunya virus detection in naturally infected *Aedes aegypti* in Ecuador. Acta Trop. 2018 Jan 1; 177:74–80.
- Ministerio de Salud Pública del Ecuador. Enfermedades Transmitidas por Vectores Ecuador, SE 52 / 2018 [Internet]. 2019. Available from: https://www.salud.gob.ec/wp-content/uploads/2019/01/ GACETA-VECTORES-SE-52.pdf.
- Campos SS, Fernandes RS, Dos Santos AAC, De Miranda RM, Telleria EL, Ferreira-De-Brito A, et al. Zika virus can be venereally transmitted between *Aedes aegypti* mosquitoes. Parasit Vectors. 2017 Dec 15; 10(1).
- 30. Thangamani S, Huang J, Hart CE, Guzman H, Tesh RB. Vertical transmission of Zika virus in *Aedes aegypti* mosquitoes. Am J Trop Med Hyg. 2016 Nov 1; 95(5):1169–73.
- **31.** Costa CF da, Silva AV da, Nascimento VA do, Souza VC de, Monteiro DC da S, Terrazas WCM, et al. Evidence of vertical transmission of Zika virus in field-collected eggs of *Aedes aegypti* in the Brazilian Amazon. PLoS Negl Trop Dis. 2018 Jul 16; 12(7).
- **32.** Ferreira-De-Brito A, Ribeiro IP, De Miranda RM, Fernandes RS, Campos SS, Da Silva KAB, et al. First detection of natural infection of *Aedes aegypti* with Zika virus in Brazil and throughout South America. Mem Inst Oswaldo Cruz. 2016 Oct 1; 111(10):655–8.
- 33. Calle-Tobón A, Pérez-Pérez J, Rojo R, Rojas-Montoya W, Triana-Chavez O, Rúa-Uribe G, et al. Surveillance of Zika virus in field-caught *Aedes aegypti* and *Aedes albopictus* suggests important role of male mosquitoes in viral populations maintenance in Medellín, Colombia. Infect Genet Evol. 2020 Nov 1: 85.
- Lequime S, Paul RE, Lambrechts L. Determinants of Arbovirus Vertical Transmission in Mosquitoes. Vol. 12, PLoS Pathogens. Public Library of Science; 2016. p. e1005548.
- 35. Maniero VC, Rangel PSC, Coelho LMC, Silva CSB, Aguiar RS, Lamas CC, et al. Identification of Zika virus in immature phases of *Aedes aegypti* and *Aedes albopictus*: A surveillance strategy for outbreak anticipation. Brazilian J Med Biol Res. 2019; 52(11).
- Ciota AT, Bialosuknia SM, Ehrbar DJ, Kramer LD. Vertical transmission of Zika virus by Aedes aegypti and Ae. albopictus mosquitoes. Vol. 23, Emerging Infectious Diseases. Centers for Disease Control and Prevention (CDC); 2017. p. 880–2.
- **37.** Reese SM, Mossel EC, Beaty MK, Beck ET, Geske D, Blair CD, et al. Identification of super-infected *Aedes triseriatus* mosquitoes collected as eggs from the field and partial characterization of the infecting la Crosse viruses. Virol J. 2010; 7.
- Ribeiro GS, Hamer GL, Diallo M, Kitron U, Ko AI, Weaver SC. Influence of herd immunity in the cyclical nature of arboviruses. Curr Opin Virol. 2020 Feb 1; 40:1–10. <u>https://doi.org/10.1016/j.coviro.2020.02</u>. 004 PMID: 32193135
- Netto EM, Moreira-Soto A, Pedroso C, Höser C, Funk S, Kucharski AJ, et al. High Zika Virus Seroprevalence in Salvador, Northeastern Brazil Limits the Potential for Further Outbreaks. MBio. 2017 Nov 1; 8 (6). https://doi.org/10.1128/mBio.01390-17 PMID: 29138300